Author's response to reviews

Title: Treatment of hepatic encephalopathy by on-line hemodiafiltration: A case series study

Authors:
  Shinju Arata (s_arata@yokohama-cu.ac.jp)
  Katsuaki Tanaka (k_tanaka@urahp.yokohama-cu.ac.jp)
  Kazuhisa Takayama (k-takayama@star.ocn.ne.jp)
  Yoshihiro Moriwaki (qgc3@urahp.yokohama-cu.ac.jp)
  Noriyuki Suzuki (yqsuzuki@urahp.yokohama-cu.ac.jp)
  Mitsugi Sugiyama (sugiyama@urahp.yokohama-cu.ac.jp)
  Kazuo Aoyagi (urame@urahp.yokohama-cu.ac.jp)

Version: 5 Date: 11 May 2010

Author's response to reviews:

May 11, 2010
Dear Editor of BMC Emergency Medicine:
Object: MS: 1214835103063241
Treatment of hepatic encephalopathy by on-line hemodiafiltration: A case series study
We have reviewed the above manuscript and made changes according to your comments.

Formatting changes to be made:
---------------------------------
Major revisions (we require the author to make these changes)

Figures: All figures must be cited in the text; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Figure 5 does not appear to be cited. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9. Figures also need to be cited in numerical order.

Figure 5 in the results section of the text now appears in page 10.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: Please ensure all mention of figures is consistent. It can be ?Figure? or ?Fig.?, but not a mixture of both.
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.
We have unified it “Figure”. And we have minimized figures.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have changed “Case 16” to “Case 17” in page 10. We also revised a small letter for a capital letter in Table 1.

Sincerely yours,
Shinju Arata